Author's response to reviews

Title: A comparison of specialist rehabilitation and care assistant support with specialist rehabilitation alone and usual care for people with Parkinson’s living in the community: study protocol for a randomised controlled trial.

Authors:

Heather Gage (h.gage@surrey.ac.uk)
Sharlene Ting (s.ting@surrey.ac.uk)
Peter Williams (p.williams@surrey.ac.uk)
Karen Bryan (k.bryan@surrey.ac.uk)
Julie Kaye (juliemkaye@yahoo.co.uk)
Beverly Castleton (beverly.castleton@surreypct.nhs.uk)
Patrick Trend (ptrend@royalsurrey.nhs.uk)
Derrick Wade (derrick.wade@noc.nhs.uk)

Version: 7 Date: 18 November 2011

Author’s response to reviews: see over
Dear Editors,

We have amended the manuscript in the light of the formatting requests.

- Indicated that all authors have read and approved the final manuscript
- Replace reference 17 and 18 with ones with functioning weblinks
- Put the Figure 1 legend after the references
- Shown in the text where Figure 1 occurs

All changes are shown in red.

Yours sincerely

Karen Bryan
Corresponding Author